Author’s response to reviews

Title: Mopane worm allergy: A Case Report

Authors:

okechukwu A okezie (okeyokezie@yahoo.com)
koloi k kgomotso (kkoloi@bhp.org.bw)
Mavis M letswiti (mletswiti@bhp.org.bw)

Version: 9 Date: 21 August 2009

Author’s response to reviews:

The manuscript revision elaborated more on food allergy and intolerance. It also expanded on prevalence and standards in the management of food intolerance and allergy.

The revision also compared recommended standard with our practice in Botswana exposing how management of allergy is structured in consideration of resource limitations and acceptable practice within Botswana context.